Author's response to reviews

Title: Associations between sexual behaviour change among young people and decline in HIV prevalence in Zambia

Authors:

    Ingvild F Sandoy (ingvild.sandoy@cih.uib.no)
    Charles Michelo (charles.michelo@student.uib.no)
    Seter Siziya (ssiziya@yahoo.com)
    Knut Fylkesnes (knut.fylkesnes@cih.uib.no)

Version: 8 Date: 19 April 2007

Author's response to reviews:

All the suggested formatting changes have been made and we have checked all the tables and text and corrected a couple of minor errors.